\begin{thebibliography}{10}

\bibitem{Aho74}
Alfred~V. Aho, John~E. Hopcroft, and Jeffrey~D. Ullman.
\newblock {\em The Design and Analysis of Computer Algorithms}.
\newblock Addison-Wesley, 1974.

\bibitem{SML-NJ89}
Andrew~W. Appel and David~B. MacQueen.
\newblock {\em {Standard ML Reference Manual}}.
\newblock {Princeton University and AT\&T Bell Laboratories}, {Preliminary}
  edition, 1989.
\newblock {Distributed along with the Standard ML of New Jersey Compiler}.

\bibitem{Clement86}
D.~Cl\'{e}ment, J.~Despeyroux, T.~Despeyroux, and G.~Kahn.
\newblock {A Simple Applicative Language: Mini-ML}.
\newblock In {\em {Proceedings of the ACM Symposium on LISP and Functional
  Programming}}, pages {13--27}, {August} 1986.

\bibitem{Damas82}
L.~Damas and R.~Milner.
\newblock {Principle Type Schemes for Functional Programs}.
\newblock In {\em {Proceedings of the 9th ACM Symposium on Principles of
  Programming Languages}}, pages {207--212}, 1982.

\bibitem{Gupta90}
Shail~Aditya Gupta.
\newblock {An Incremental Type Inference System for the Programming Language
  Id}.
\newblock Technical Report MIT/LCS/TR-488, Laboratory for Computer Science, 545
  Technology Square, MIT, Cambridge, MA 02139, November 1990.
\newblock {First published as the author's Master's thesis}.

\bibitem{Mairson90}
Harry~G. Mairson.
\newblock {Deciding ML Typability is Complete for Deterministic Exponential
  Time}.
\newblock In {\em {Proceedings of the 17th ACM Symposium on Principles of
  Programming Languages}}, pages 382--401, {January} 1990.

\bibitem{Meertens83}
Lambert Meertens.
\newblock {Incremental Polymorphic Type Checking in B}.
\newblock In {\em {Proceedings of the 10th ACM Symposium on Principles of
  Programming Languages}}, {January} 1983.

\bibitem{Milner78}
Robin Milner.
\newblock {A Theory of Type Polymorphism in Programming}.
\newblock {\em Journal of Computer and System Sciences}, 17:348--375, 1978.

\bibitem{Milner90}
Robin Milner, Mads Tofte, and Robert Harper.
\newblock {\em {The Definition of Standard ML}}.
\newblock {The MIT Press, Cambridge, Massachusetts}, 1990.

\bibitem{Nikhil85}
Rishiyur~S. Nikhil.
\newblock {Practical Polymorphism}.
\newblock In {\em Proceedings of the Conference on Functional Programming
  Languages and Computer Architectures, Nancy, FRANCE}, volume 201 of {\em
  Lecture notes in Computer Science}. Springer-Verlag, September 1985.

\bibitem{Id90}
Rishiyur~S. Nikhil.
\newblock {Id Version 90.0 Reference Manual}.
\newblock Technical Report CSG Memo 284-1, MIT Laboratory for Computer Science,
  545 Technology Square, Cambridge, MA 02139, July 1990.

\bibitem{Idworld}
Rishiyur~S. Nikhil, P.~R. Fenstermacher, J.~E. Hicks, and R.~P. Johnson.
\newblock {\em {Id World Reference Manual}}.
\newblock Computation Structures Group, MIT Laboratory for Computer Science,
  545 Technology Square, Cambridge, MA 02139, revised edition, November 1989.

\bibitem{Tofte88}
Mads Tofte.
\newblock {\em {Operational Semantics and Polymorphic Type Inference}}.
\newblock PhD thesis, {University of Edinburgh, Department of Computer
  Science}, 1988.
\newblock {Also published as ECS-LFCS-88-54}.

\bibitem{Toyn87}
Ian Toyn, Alan Dix, and Colin Runciman.
\newblock {Performance Polymorphism}.
\newblock In {\em {Functional Programming Languages and Computer
  Architecture}}, volume 274 of {\em {Lecture Notes in Computer Science}},
  pages 325--346. {Springer-Verlag}, 1987.
\newblock {Proceedings of the FPCA Conference held in Portland, Oregon, 1987.}

\bibitem{Turner85}
David~A. Turner.
\newblock {Miranda: A non-strict functional language with polymorphic types}.
\newblock In {\em Proceedings of the Conference on Functional Programming
  Languages and Computer Architectures, Nancy, FRANCE}, volume 201 of {\em
  Lecture Notes in Computer Science}. Springer-Verlag, September 1985.

\bibitem{Turner86}
David~A. Turner.
\newblock {An Overview of Miranda}.
\newblock {\em {SIGPLAN Notices}}, 21(12):{158--166}, December 1986.

\end{thebibliography}

% -*- Mode:TeX -*-

\section{Background}
\label{background}

Nikhil in \cite{Nikhil85} pointed out the disparity between the
goals of an incremental programming environment, and
ML-like type inference system. ML \cite{SML-NJ89,Milner90} is
interactive, but a session in ML is essentially a large lexically nested
ML program. Each toplevel definition has the rest of the session as its
scope. Thus, editing an earlier definition may force the user to recompile
and reload all the intermediate definitions that used
it\footnote{SML's module mechanism gives some relief in this
respect, due to separate compilation.}. In building
large systems, the recompilation and reloading of large pieces of potentially
unrelated code, just to recreate the same environment every time a small
error is detected, is at best, quite annoying.

In Miranda \cite{Turner85,Turner86}, this problem is resolved by making the
unit of compilation to be a whole file (also called a ``Miranda script'').
Definitions within a file may appear in any order and the compiler is
responsible for reordering them during compilation.  The interactive
session only evaluates expressions using definitions from the current
script. But, definition level incrementality has been lost, and editing
forces entire files to be recompiled.

In contrast, Lisp programming environments smoothly integrate the
editor, the compiler, and the read-evaluate-print loop. The unit of
compilation in these systems is a single top-level definition. They
allow the user to furnish multiple top-level definitions, either
together, or one by one in any order, resolving global references to
other definitions automatically by dynamic linking.  The user can
test, debug, and edit these definitions incrementally, without waiting
to write the complete program or having to recompile a substantial fraction
of the definitions already supplied.

In Id, we attempt to achieve the flexibility of Lisp-like environments
along with static type-checking.  The programming environment for Id,
called ``Id World''\cite{Idworld}, smoothly integrates the editor, the Id
compiler, and the underlying execution vehicle. Like Lisp, our unit of
compilation is a single top-level definition. Each compiled definition is
accumulated into a flat global environment, implying that edited
definitions are immediately and automatically made available to other
definitions that use them.  Simple interprocedural book-keeping maintains
type-consistency between the definition and the uses of each top-level
identifier.  The book-keeping mechanism derives flexibility from the
polymorphism of the definitions, flagging only inconsistent definitions for
recompilation. As a consequence, we permit out of order compilation,
redefinition, and editing with minimum overhead, while still guaranteeing
type correctness before program execution.  We also allow executing
partially defined Id programs, where the undefined identifiers simply
generate an exception if actually used at run-time.  Again, the incremental
book-keeping mechanism helps in incorporating the missing parts as
and when they become available with minimum overhead.
